



# **Modeling Radio Wave Propagation for Wireless Sensor Networks in Vegetated Environments: A Systematic Literature Review**

Alexis Barrios-Ulloa <sup>1,2,\*</sup>, Paola Patricia Ariza-Colpas <sup>1</sup>, Hernando Sánchez-Moreno <sup>3</sup>, Alejandra Paola Quintero-Linero <sup>4</sup> and Emiro De la Hoz-Franco <sup>2,\*</sup>

- <sup>1</sup> Department of Computer Science and Electronics, Faculty of Engineering, Universidad de la Costa, Barranquilla 080002, Colombia; pariza1@cuc.edu.co
- <sup>2</sup> Department of Electronics Engineering, Faculty of Engineering, Universidad de Sucre, Sincelejo 700001, Colombia
- <sup>3</sup> Facultad de Ciencias Básicas, Universidad Simón Bolívar, Centro de Investigación e Innovación en CienciasMarinas y Limnológicas del Caribe Colombiano "CICMAR", Barranquilla 080001, Colombia; hsanchez13@unisimonbolivar.edu.co
- <sup>4</sup> Microbiology Program, Universidad Popular del Cesar, Valledupar 200002, Colombia; alejandraquinterol@unicesar.edu.co
- \* Correspondence: abarrios56@cuc.edu.co or alexis.barrios@unisucre.edu.co (A.B.-U.); edelahoz@cuc.edu.co (E.D.I.H.-F.)

check for **updates** 

Citation: Barrios-Ulloa, A.; Ariza-Colpas, P.P.; Sánchez-Moreno, H.; Quintero-Linero, A.P.; De la Hoz-Franco, E. Modeling Radio Wave Propagation for Wireless Sensor Networks in Vegetated Environments: A Systematic Literature Review. *Sensors* 2022, 22, 5285. https:// doi.org/10.3390/s22145285

Academic Editors: Davide Brunelli, Francesca Cuomo, Luca Catarinucci, Gregor Klancar and Matteo Nardello

Received: 30 March 2022 Accepted: 27 May 2022 Published: 15 July 2022

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). Abstract: The use of wireless sensor networks (WSN) for monitoring variables in agricultural environments and natural forests has been increasing in recent years. However, the sizing of these systems is affected by the inaccuracy of the radio wave propagation models used, leading to possible increased costs and measurement errors. This systematic literature review (SLR) aims to identify propagation models widely used in WSN deployments in agricultural or naturally vegetated environments and their effectiveness in estimating signal losses. We also identified today's wireless technologies most used in precision agriculture (PA) system implementations. In addition, the results of studies focused on the development of new propagation models for different environments are evaluated. Scientific and technical analysis is presented based on articles consulted in different specialized databases, which were selected according to different combinations of criteria. The results show that, in most of the application cases, vegetative models present high error values when estimating attenuation.

**Keywords:** attenuation; vegetated environments; propagation models; path loss; systematic revision of literature; wireless technologies; WSN

## 1. Introduction

Wireless sensor networks (WSNs) based on the Internet of Things (IoT) are widely used in different scenarios, including agricultural or with the presence of natural vegetation. These tools are used to obtain the values of the different variables of a plantation or forest, whether related to climate, soil, or plants. They are one of the most promising data collection methods in agricultural applications due to their low cost and improvements in environmental monitoring [1,2]. This information is useful for the personnel in charge of crop management as it facilitates the management of the processes in their plantations [3,4]. This form of crop management is called precision agriculture (PA) and allows managers to more efficiently manage both the facilities and the inputs required by the plantation. It is important to highlight that the monitoring of variables in PA contexts has a positive impact on irrigation management, reduction in environmental impact, and fertilizer management, among others [5,6].

The number of nodes that can be deployed and interconnected ranges from one or two units to thousands, depending on geographical coverage and topography [5], which is enhanced by the mMTC (Massive Machine-Type Communication) scenario of 5G [7,8].

The correct deployment of the nodes in the field is based on the efficient management of the transmitter power, knowing beforehand the signal attenuation through an adequate propagation model that leads to adequate data collection [9–11].

The wireless signal is often obstructed in its path by trees, hills, mountains, and buildings, among others [12,13], resulting in phenomena such as scattering, diffraction, and reflection [14] (see Figure 1). Even the signal strength in line-of-sight (LOS) links and static transceivers can be altered due to scatterers or moving objects affecting the propagation environment [15]. For this reason and given the importance of WSN implementation in PA contexts, it is necessary to bring propagation model estimates closer to observed values as they are closely related to WSN performance prediction metrics (energy loss, path optimization, and link reliability) [15,16].



Figure 1. Examples of propagation modes in vegetation. Source: Refs. [17,18].

Propagation models that estimate path loss or characterize attenuation in wireless links [19] play an important role in the deployment of WSNs in different large-scale contexts [20]. As for its application in PA, it is used to model the attenuation through the foliage, estimating metrics useful in planning the deployment of WSNs. In the scientific literature, they are grouped into three main categories: empirical, deterministic, and stochastic [21]. However, it is important to note that this classification cannot be considered accurate as some models can be classified in more than one category without being mutually exclusive. The empirical ones are obtained from rigorous field measurements in addition to the analysis of the statistical characteristics describing the behavior of the signal using an equation [22]. Deterministic ones are based on the study of propagation phenomena, as well as on the knowledge of the environment in which the network is deployed; they can be simple and range from only including the distance separating the nodes to considering more complex factors in its development (e.g., multipath fading) [23]. Finally, stochastics result from the modeling of the environment based on random variables [24], which generates high error values in the results.

There are different propagation models oriented to WSNs. However, many of them have high error values or are not very rigorous because they do not consider relevant factors in their equations (e.g., antenna height).

The main objective of this review work is to determine which propagation models are used in the characterization of signal losses of WSNs deployed in agricultural or vegetation environments, as well as their efficiency in different scenarios (forest, outdoor crop, greenhouse, meadow). The specific objectives of this systematic literature review (SLR) are:

 Contextualize the use of propagation models used in WSN planning in PA and vegetated environments.

- Identify applications in vegetation scenarios where propagation studies associated with WSN deployment have been conducted.
- Identify the wireless technologies most used in propagation studies oriented to the deployment of WSNs in agricultural or vegetation environments.

The main motivation of this study is to present an SLR that identifies the most widely used models with the greatest impact on WSNs in agricultural contexts or environments with similar characteristics.

The contributions of this document are specified below:

- Effectiveness of propagation models in WSN applications related to agriculture or forestry environments.
- Identification of the predominant wireless technologies in propagation studies for sizing WSN nodes in agriculture and similar scenarios.
- Identification of the most used techniques to validate propagation models in vegetated environments.

The structure of this article is as follows: Section 2 presents the related works, Section 3 presents the methodology used during the review, Section 4 describes the scientometric analysis, and Section 5 presents the detailed technical analysis. Sections 6 and 7 present, respectively, the discussion and the results obtained. Finally, Sections 8 and 9 present the conclusions, recommendations, and future work.

## 2. Related Works

Regardless of the type or way of obtaining each propagation model (also called path loss models), we must be aware of the particularities of the scenario where the WSN is deployed. Therefore, it is normal to (a) consider the models validated and ready to be applied, (b) adjust these models, and (c) develop a new one that allows to adequately characterize the attenuation.

After searching specialized databases, the number of literature review articles on WSNoriented radio wave propagation models is minimal, let alone focused on their applications in agricultural contexts. In Ref. [15], they reviewed and compared models, describing their limitations in WSNs in general, without focusing on specific scenarios. In Ref. [25], they evaluated different models frequently used in WSN planning through a simulation platform. They demonstrated the usefulness of modeling the network in vegetation environments because their results significantly influence network performance. In Ref. [24], they studied different path loss prediction methods, not specifically intended for WSNs, and presented a new taxonomy of models based on the similarities and differences between them.

Other works, although not of review, evaluate propagation models intended for WSNs in agricultural or similar scenarios of forests or with the presence of pastures. In Ref. [26], they contrasted several empirical models in different bands and crops, relating received signal strength (*RSSI*) values to path loss data. In Ref. [27], the foliage attenuation model recommended by the International Telecommunication Union (ITU-R), the Modified Exponential Decay model (MED), and Weissberger model were simulated and analyzed to predict the connectivity as well as the maximum coverage of the nodes within the communication path. In Ref. [3], they described the effects of distance, antenna height, and vegetation depth on the models. However, these papers focus on specific issues and do not present a comprehensive systematic review on aspects of vegetation propagation modeling for WSNs, such as: the effectiveness of the most used models, study scenarios, or analysis of validation results.

The initial information search shows that the number of SLR or bibliometric review articles on modeling radio wave propagation in WSNs deployed in vegetated environments (e.g., agricultural fields) is small. On the contrary, numerous reviews are focusing, for example, on sensors [28], optimization models [29], WSN applications in PA [30], and the use of communication technologies from frequency bands, data rate, and transmission range, among other aspects [31]. Therefore, researchers new to propagation modeling in vegetated WSN systems do not have a wide variety of review literature at their disposal

and must devote part of their working time to gathering a wealth of information due to many scenarios under study, the available wireless technologies, and the lack of unified methods to validate the results. In that sense, this SLR article will be a valuable tool for consultation and support on the subject, not only for the identification of the most used models, scenarios, and technologies in the research on the modeling of radio signal propagation in agricultural WSN systems or similar but also the evaluation and analysis of the effectiveness of these models from the statistical methods used as a validation tool.

## 3. Methodology

Within the field of research, SLRs have become an important contribution to the scientific community since they are articles that compile, summarize, criticize, and synthesize the information available on a specific topic. This study considers the SLR recommendations presented in Refs. [32–35]. The structure and methodology of the SLR articles used in Ref. [36] were considered, where they established search criteria based on keywords found in previous reviews and exclusion principles that allowed the information found to be purified.

In the field of WSNs, they defined in Ref. [5] a set of questions to be solved in the implemented SLR. They performed an information search in specialized databases and created search strings from keywords. The selection criteria considered the period of publication (in years), as well as the selection of articles included in journals indexed in the Science Citation Index (SCI). In Ref. [30], they established an SLR process to identify applications and criteria in PA technology selection. Based on these guiding references, SLRs' work on radio wave propagation models in agricultural environments was developed in three phases: (1) design of the search string, (2) application of the search string and data collection, and (3) analysis of the results.

In phase 1, a series of questions were posed:

Q1. What are the most used propagation models for sizing WSNs in agriculture?

Q2. In which crops, or scenarios with vegetation, has propagation modeling work been done for WSN?

Q3. What are the communication technologies used as a tool for modeling attenuation in WSNs in agricultural or similar environments?

Subsequently, the search strings were structured based on the compilation of the information presented in review and overviews articles. Figure 2 shows the conceptual map considered in its construction.



Figure 2. Conceptual map used in the construction of search strings.

Table 1 presents the thematic axes and explains the strategies for combining these criteria in the elaboration of search strings. The combinations were created in ascending and descending order. In addition, the terms used in the exclusion criteria are presented.

Table 1. Thematic axes and combination strategies used in the review.

Thematic Axis 1	Thematic Axis 2	Thematic Axis 3	Exclusion Criteria			
Agriculture	WSN	Path loss	Algorithm			
Crop		Path loss model	Image			
Farm		Wave propagation	Indoor			
Vegetation			Irrigation			
0			Localization			
			Monitoring			
			Solar energy			
	Combi	inations				
Link 1:		Axis 1 AND Axis 2				
Link 2:	Axis 1 ANI	O Axis 2 AND NOT Exclu	sion criteria			
Link 3:	Axis 1 AND Axis 2 AND Axis 3					
Link 4:	Axis 1 AND Axis 2 AND Axis 3 NOT Exclusion criteria					
Link 5:	Axis 2 AND Axis 3					
Link 6:	Axis 2 AND Axis 3 NOT Exclusion criteria					

In phase 2, search strings were applied in five highly recognized specialized databases: Scopus, Web of Sciences (WoS), Institute of Electrical and Electronics Engineers (IEEE), Science Direct, and Springer. Other sources of information, such as Google Scholar and ResearchGate, were not considered. On these platforms, there may be cases where researchers link documents that do not correspond to scientific articles. In addition, they do not apply the same rigorous publication standards that apply to specialized databases.

Different terms were used to establish exclusion criteria. The concepts "algorithm", "image", "irrigation", "monitoring", and "solar energy" were associated with results in which the main topic was the implementation of WSNs for monitoring, so they were not considered. Excluded as well were those articles in which the concepts "indoor" and "localization" could be related to the topic of propagation, but most of them did not contain aspects of WSNs, agriculture, or vegetation. The selected articles were published between 1 January 2011 and 31 December 2021. Those chains that did not yield results, or where the results were not related to this SLR work, they were discarded.

The initial selection process of the articles was based on the analysis of the title, keywords, abstract, and conclusions in order to establish their coherence with the main objective of the research. Other exclusion criteria were language, publication status, and journal classification. Only papers submitted in English were selected, and those in preprint that had not undergone peer review were discarded. In terms of categorization, the selection of journal articles was limited to those ranked between the first and fourth quartile (Q1–Q4) in the Scimago Journal Report (SJR) and the Journal Citation Report (JCR), eliminating uncategorized articles. The latter criterion was not applied to the documents corresponding to the conferences. In the same phase, to strengthen the search, the review included the references presented in the selected papers, adding 14 additional papers. The existence of repeated documents in two or more databases was reviewed in detail, and 69 out of the total number of articles chosen were eliminated and counted only once.

Initially, 138 articles related to the topic of propagation models for WSNs in agricultural or vegetation environments were selected and analyzed. In this phase, the most relevant references were identified, creating a database with scientific and technical information. Subsequently, a more detailed analysis was carried out, reviewing in each article elements such as problems addressed, methodology used, the technology employed, and problems to be solved, which made it possible to reduce the number of articles to 84. The scientometric and technical variables of each were documented. As an additional criterion for the final selection, the information provided by the authors regarding the measurement or

simulation environments, the methodology used, and the presentation of the results was taken into account.

Finally, in phase 3, the results were analyzed based on quantitative and qualitative information, correlating the different categories examined in the review process.

#### 4. Scientometric Analysis

For each of the 84 references, the following scientometric information was quantified and recorded: name of the journal or conference, quartile ranking in Scopus and WoS, nationality of the journal in which it was published, country of origin of the work, and authors' institution of affiliation. Figure 3 shows the number of articles related to propagation models, WSN, and agriculture.





It is evident that the number of publications on this specific topic demonstrates its relevance within propagation and WSN research, mainly in works related to (1) characterization of attenuation in environments with the presence of vegetation and evaluation of widely recognized models, and (2) development of new propagation models associated with agricultural or vegetated environments. Figure 3 shows that the maximum peak of publications was reached in 2019, before the start of the pandemic caused by the SARS-CoV-2 virus; this situation caused a large part of the research resources to be allocated to the study of aspects related to the pandemic.

Figure 4 shows that 50% of the selected papers were published in journals, while the remaining 50% corresponded to conferences. These indicators, as well as the large percentage of publications, ranked in Q1 (see Figure 4a), demonstrate the relevance of this topic in the field of research.





The large number of countries that have journals or hold conferences dedicated to the subject demonstrates the interest and importance that propagation studies generate within the research community, especially regarding improving the layout of sensor nodes in scenarios where vegetation affects the signal quality of communications links. Figure 5 shows that the countries with the highest number of articles received are the United States, Netherlands, Switzerland, and China, which have five or more publications in their different journals or conference proceedings.





Figure 6 presents information on the countries with the largest number of papers related to this SLR. The list is headed by China, followed by India, Malaysia, Japan, the United States, and Spain, nations in which agriculture is of great importance as an economic activity due to their growing population and because they dedicate part of their production to exports. In addition, the information in the figure shows that there is a general interest in much of the world for projects related to research activities in PA, WSN, and propagation. As an evaluation criterion for this item, relevance was given to the country of affiliation of the authors, and, in cases where the origin of more than one country was recorded, the origin of the first author was chosen.



Figure 6. Number of publications generated by country.

The journals with the highest number of publications related to the topic of this SLR are presented in Table 2. The number of articles, as well as the categorization of the journals according to quartile in SJR and JCR, show that the topic of radio signal propagation aspects in WSNs deployed in agricultural or similar fields is of great importance in the global scientific context.

Publishing	Number of Posts	Country	Туре	ISSN	Quartile in SJR	Quartile in JCR
Computers and Electronics in Agriculture	9	Netherlands	Journal	1681699	Q1	Q1
Sensors	5	Switzerland	Journal	14243210	Q2	Q1
Progress in Electromagnetics Research C	3	USA	Journal	19378718	Q3	Q3
Wireless Personal Communications	3	Netherlands	Journal	1572834X	Q3	Q4

Table 2. Journals with the most publications.

## 5. Technical Analysis

When examining research that aims to characterize propagation in agricultural or similar environments, different technical aspects are considered. The most relevant considerations resulting from this SLR are presented below from three approaches: radio wave propagation models for WSN in agricultural or vegetated scenarios, application scenarios for propagation studies, and technologies used (wireless, frequency bands, and tools).

## 5.1. Radiowave Propagation Models for WSNs in Agricultural or Vegetated Environments

The path of radio waves in forest, crops, or similar environments is affected by numerous parameters: the vegetation path length (*d*), the average tree height (*hv*), the height of the receiving antenna (*hr*), the path elevation angle ( $\theta$ ), and the distance from the antenna to the road forest (*dw*) [37] (see Figure 7). In addition, each crop and vegetated environment has its propagation characteristics, so many studies aimed at characterizing radio signal attenuation in the wireless channel are based on recognized models. This occurs when the network must be planned and sized, or when the effectiveness of the proposed new models must be evaluated.



**Figure 7.** Parameters considered in the propagation path of the radio wave in vegetation. Source: Ref. [37].

As a result of this SLR, it became evident that, in agricultural or forestry scenarios, empirical models are the most used in the planning and installation of WSNs, as shown in Figure 8.

Due to their high efficiency in loss estimation, in addition to their low mathematical complexity, empirical models are highly appreciated in wireless network dimensioning. Among them is the MED, one of the most widely used due to the high levels of accuracy it offers when determining canopy losses [38]. The Equation (1) describing it is presented below [39]:

$$Att_{MED} = x f^y d^z \tag{1}$$

where f corresponds to the frequency given in MHz, and d is the depth of the vegetation in meters. In addition, x, y, and z are parameters that must be tuned from the measurements made in each scenario where their use is required [40]. In Ref. [27], it was contrasted,

along with other models, to establish which one provided greater effectiveness and, thus, adequately sized WSNs in different environments, including agriculture. MED was also applied in Refs. [38,39] for comparison with a model proposed by the authors.



Figure 8. Types of models identified in the SLR.

Different models have emerged from the MED, such as Weissberger, ITU-R, adjusted ITU-R (FITU-R), and the COoperation in science and technology (COST-235), which have been widely reported in the literature. In the selected papers in this SLR, the ITU-R model was the most used in the estimation of radio signal power loss in vegetated environments where they implemented or simulated WSNs. The Equation (2) representing the losses of this model is presented below [12]:

$$PL_{ITU-R}(dB) = 0.2 \times f^{0.3} \times d^{0.6}.$$
 (2)

where *f* is the frequency in MHz and *d* corresponds to the distance. The model was used in Ref. [41] in the calculation of signal losses in different types of plantations, in addition to different variations of garden scenarios (coconut trees with green grass, dry open grass, and wet green grass), and, in Ref. [26], a comparison was made concerning the measurements made in each of these scenarios. In Refs. [3,27], they presented models that were examined to determine which best represented the losses in the proposed scenarios. However, in these cases, they were compared with simulations and not with actual measurements. In Refs. [12,18,42–44], the ITU-R estimates were compared with the results obtained when developing new models to calculate the effectiveness of the models.

Another widely used model within the set of articles examined in this SLR is COST-235, which also comes from adjustments made to the MED. Equation (3) represents it [45]:

$$C_L(dB) = \begin{cases} 15.6 \times f^{-0.009} \times d^{0.26}, in \, leaf\\ 26.6 \times f^{-0.2} \times d^{0.5}, \, out \, of \, leaf. \end{cases}$$
(3)

In this equation, f is the frequency value in MHz and d is the distance in meters. Like the models described above, COST-235 was also used to identify a suitable method describing foliage losses in crop or forestry scenarios, as is the case reported in Ref. [46], where the impact of the orientation of WSN nodes was also analyzed.

Like ITU-R and COST-235, the Weissberger model is derived from the MED and has been widely used in the work that is part of this SLR. Equation (4) describes the losses of this model [47]:

$$L_{\rm (dB)} = \begin{cases} 1.33 \times f^{0.284} \times d^{0.588}; si \ 14 < d \le 400\\ 0.45 \times f^{0.284} \times d; si \ 0 < d < 14 \end{cases}$$
(4)

In this case, the frequency (f) is given in GHz and d in meters. This model is only defined at a foliage depth of up to 400 m.

The FITU-R model is also derived from the MED and is one of the most widely used in the articles considered in this work. It was proposed for situations where the transmitting or receiving antenna is located near a small strip of trees so that most of the signal propagates through them. Equation (5) represents it [3]:

$$L_{FITU-R(dB)} = \begin{cases} 0.37 \times f^{-0.18} \times d^{0.54}, & out \ of \ leaf\\ 0.39 \times f^{-0.39} \times d^{0.25}, & in \ leaf \end{cases}$$
(5)

where *f* is expressed in MHz and *d* in meters.

Although empirical models are the most used in the works considered in this review, the use or development of deterministic models is also reported, as is the case of the Two-Ray model. It considers the effects of direct and reflected waves to the ground on signal propagation [48]. Equation (6) describes the losses of this model [49]:

$$P_{LTwo}(dB) = 40\log(d) - 20\log(h_t) - 20\log(h_r) - 10\log(G_t) - 10\log(G_r)$$
(6)

where *ht* and *hr* are the heights in meters of the transmitting and receiving antennas, respectively, and *Gt* and *Gr* are the gains of the transmitting and receiving antennas in dBi.

Although Two-Ray is one of the most widely used in WSN propagation studies in PA contexts, most papers report its use in comparison to other models to estimate its effectiveness, as presented in Refs. [50–53]. Furthermore, in Refs. [16,54], it was used in the characterization of WSN radio wave propagation in fields or scenarios with the presence of vegetation. In the SLR performed in Ref. [15], it was considered within the group of most important propagation models in the estimation of signal losses in WSN in agricultural environments or where vegetation is an important factor in the propagation behavior.

Regarding the development of deterministic models, only one case is reported in Ref. [55], where they considered the effects of corn and potato field vegetation on propagation by simulating the dielectric properties of these crops.

Concerning stochastic models, they usually present high error values in the estimation of signal loss in a wireless channel. Despite this, there are reports of papers that considered its use. In Ref. [56], the authors proposed a new model for WSN applied to forest environments using the Green Obadait and other models in evaluating the effectiveness of their proposal. The use of the One-Slope model was also reported in research relating WSNs and PA. In Ref. [57], it was used to validate the use of Xbee modules in agricultural environments; while the SLR presented in Ref. [15] considered it one of the most useful in characterizing signal power loss in WSNs.

Within the wide range of radio wave propagation models that appear in the literature, it is evident that one of the most widely used is the Free Space Loss Model (FSPL). Ideal propagation conditions are assumed here without considering the phenomena of diffraction, reflection, refraction, and absorption. Moreover, it is only valid in the far-field region [58]. However, although it is a general-purpose model, it is very useful in studies of radio signal wave propagation in agricultural plantations or vegetated environments, and as with the Two-Ray, it serves as a complement to the MED or its derivations, such as ITU-R, FITU-R, and COST-235, because the losses predicted by these models are concentrated in the foliage and not in the entire path. Therefore, FSPL losses are usually summed to calculate path losses [15]. Equation (7) describes the losses in free space, where the frequency (f) is expressed in MHz and the distance from the receiving antenna to the transmitting antenna d is given in kilometers [59]:

$$PL(dB) = 32.44 + 20\log d + 20\log f \tag{7}$$

Figure 9 shows the most used models according to the articles that were subject to SLR according to the number of times they were considered in the different studies. Several papers considered more than one model in the characterization of loss in different scenarios or the comparative analysis of new proposals.

Although the models described in this section have already been validated and widely used, they generally present restrictions in their use, which have an impact on the efficiency

of loss estimation Therefore, many studies focus on the development of new propagation models. For example, in Refs. [18,43,44,56,60,61] empirical models were proposed based on the estimation of signal power losses in each of the scenarios considered in the respective papers. As documented in the widely known models, deterministic proposals are not commonly used; however, some developments are also presented in Refs. [53,62]. Table 3 contains a summary of the most important characteristics of the models commonly used in research related to radio wave propagation in vegetated environments.





Table 3. Summary of	propagation n	nodels most use	ed in the SLR articles.
---------------------	---------------	-----------------	-------------------------

Model	Vegetative Model	Antenna Height	Antenna Gain	Conditions	Observations	References
MED	Yes	No	No	<i>d</i> < 400 m 200 MHz < <i>f</i> < 95 GHz	Applicable in communication links obstructed by dense, dry, leafy trees; present in temperate latitude forests.	[27,38–40,63]
ITU-R	Yes	No	No	d < 400  m 200 MHz $< f < 95 \text{ GHz}$ For f above 1 GHz, it only considers the diffracted components around the vegetation and from above, in addition to the component reflected from the ground.	It is proposed for cases where the transmitting or receiving antenna is located near a small grove of trees, allowing most of the signal to propagate through the vegetation.	[3,12,27,40–42,46,53,64–66]
COST-235	Yes	No	No	200 MHz < f < 95 GHz	Consider the presence and absence of leaves on trees. It can be used in vegetation scenarios up to 200 meters wide.	[3,12,18,26,41,46,53,60,67,68]
FITU-R	Yes	No	No	<i>d</i> < 400 m	Consider the presence and absence of leaves on trees. Suitable for modeling the propagation of radio waves in different seasons of the year.	[3,9,53,60,63]
Weissberger	r Yes	No	No	<i>d</i> < 400 m 230 MHz < <i>f</i> < 95 GHz	Useful when the path between the transmitter and the receiver is occupied by dense vegetation consisting of trees with low humidity.	[3,12,18,26,40–42,46,53,64]

Model	Vegetative Model	Antenna Height	Antenna Gain	Conditions	Observations	References
FSPL	No	No	Yes	$d >> \lambda$	This model does not consider the mechanisms of radio wave propagation (e.g., reflection, diffraction, refraction, absorption). It is used as a reference to compare the performance of different wireless communication technologies. It is also a complement to vegetation models to calculate losses over the entire channel path.	[15,16,26,39,46,48,56,57,69–71]
Two-Ray	No	Yes	Yes	d >> ht + hr	This model considers the effects of the ground and the reflection of the ray LOS (line-of-sight). Useful for modeling propagation over long distances. It is also a complement to vegetation models to calculate losses over the entire channel path.	[15,16,45,46,48,69]

Table 3. Cont.

#### 5.2. Application Scenarios for Propagation Studies

Most of the models presented in Section 5.1 are used in environments where the presence of vegetation affects the power of the radio signal. This SLR focuses on the study of WSN radio wave propagation models in agricultural and forestry fields. Many articles have estimated the effectiveness of new or validated models in scenarios where vegetation is present but which do not necessarily correspond to crop fields. Specifically, four types of environments predominate in the works reviewed: (1) agricultural fields, (2) agricultural greenhouses, (3) lawns or pastures, and (4) forest or jungle.

As for agricultural fields, also called outdoor crops, radio wave propagation has been modeled for a considerable number of crop types. The SLR highlights that most of the work is in cereal fields, especially rice [41,50,54,72,73], corn [25,41,74], and wheat [12,75,76]. Fruit plantations, which are also part of the outdoor crops category, have been mainly: mango [39–41] and guava [39,41].

Greenhouse crops, unlike traditional outdoor cultivation, require a closed circuit and protection from the outside, which directly affects the radio signal propagation conditions. There are not many papers presented in this SLR that consider this type of scenario in greenhouse contexts. Representative cases include the simulations developed in Ref. [29], where different models are evaluated without detailing the characteristics of the plantation. Greenhouse propagation studies of tomato seedlings were conducted in Refs. [18,47].

For grass or turf crop scenarios, most research estimates signal power loss in environments with grass of considerable height, as presented in Refs. [51,52,77,78]. Short-grass scenarios represent fewer challenges from the point of view of radio wave propagation; some of these analyses are presented in Refs. [16,77,79]. A special case is presented in Ref. [80], where the calculation of foliar losses is performed in sugarcane plantations, but, due to the characteristics of the plantation, it could also be considered as tall grass.

There are also results of work on the characterization of wireless signal attenuation in forests or jungles. In this case, most of the articles do not present detailed information on the type of shrubs present in these environments. Pine forests are the most referenced, presented

in Refs. [62,81,82], as well as oak forests, also studied in Refs. [62,82]. In Refs. [43,52,56,60], it is indicated that the considered scenario corresponds to a forest, while, in Ref. [51], it was characterized as an environment of scattered trees. In this same category are classified the jungle environments, which are not extensively examined in this SLR since few studies were found on this aspect, such as Refs. [83,84]. Figure 10 shows the distribution of the environments that were considered in the articles that were part of this SLR.

■Lawn or grass ■ Outdoor crops ■ Forest or jungle ■ Greenhouse



Figure 10. Distribution of application environments for propagation characterization work.

#### 5.3. Wireless Technologies, Frequency Bands, and Tools Used

Two technical factors are important in work conducted on radio wave propagation: (a) wireless technology used in data transmission and (b) frequency band. These two aspects are important in the selection of the technological tool to be used to perform *RSSI* measurements, which is the parameter used to determine the received power at the receiver. Equation (8) is used to estimate this value [48]:

$$RSSI = 10\log P_{r(d)}.$$
(8)

where Pr(d) corresponds to the received power and d is the distance from the transmitter to the receiver. In addition, the relationship between the power at the receiver and the path loss (PL) is expressed by Equation (9) [85]:

$$P_{L(d)} = P_t + G_t + G_r - P_{r(d)}.$$
(9)

where *Pt* is the transmit power in dBm, *Gt* is the transmit antenna gain in dBi, *Gr* is the receiver antenna gain in dBi.

Different wireless technologies can be used for the transmission of data obtained in the motes that make up a WSN, and their choice is conditioned by different factors, including transmission speed, coverage, power consumption, and cost. Research on radio wave propagation in WSNs applied to the environments that have been the subject of analysis of this SLR indicates that the trend in wireless communication technology is towards the use of Zigbee, followed by the Long-Range Wide Area Network (LoRaWAN) protocol. In addition, there are reports of measurements with Bluetooth and LoRa. Table 4 presents a comparison of these standards. In addition, there are reports of measurements with WiFi and Bluetooth. Table 4 presents a comparison of these standards.

Table 4. Comparison of Bluetooth, LoRaWAN, WiFi, and Zigbee technologies. Sources: Refs. [86–89].

Technology	Frequency Bands	Speed Rate	Coverage Range (Typical)	Energy Consumption
Bluetooth	2.4 GHz	720 kbps–1 Mbps	1–10 m	Low
LoRaWAN	433 MHz, 868 MHz	250 bps–50 kbps	>10 km	Low
WiFi	2.4 GHz, 5 GHz	1.2 Mbps–54 Mbps	Hasta 100 m	High
Zigbee	870 MHz, 902–928 MHz, 2.4 GHz	20 kbps–250 kbps	10 m–1.6 km	Low

The technology most used in the research is Zigbee, a protocol based on the IEEE 802.15.4 standard. Some of its main features are low transmission speed, low power consumption, and low cost [86]. Of the spectrum usage possibilities offered by Zigbee, 2.4 GHz, which is part of the ISM (Industrial, Scientific, and Medical) band, was the most widely used in the research consulted in this SLR. In the development of new propagation models and the characterization of attenuation in the different environments presented in the previous section, most of the projects favored the use of commercial equipment. XBee devices were the most widely used in the development of new models, as presented in Refs. [38,41,42,44,46,50,52,53,61,74,75,83,90,91]. Texas Instruments transceivers were also widely used in those research studies that opted for Zigbee communication technology, being considered in Refs. [74,92]. Not all the research with Zigbee implementation worked in the 2.4 GHz band. For example, in Ref. [54], the operating frequency of the transceiver modules was 920 MHz.

Other articles report the use of WiFi in the implementation of the wireless solution. This technology is based on the IEEE 802.11 standard, and some of its features are high speed, low coverage, and reduced power consumption [93]. All the articles reporting their implementation used the 2.4 GHz band, with equipment working on the IEEE 802.11g standard [60,94], IEEE 802.11b [73,94], and IEEE 802.11n [94]. No information on the standard or working frequency was presented in Ref. [95], and no data on the equipment used was presented in Ref. [56].

Other technologies meet the conditions that allow their use in the communication stage of WSN motes, but their use is not widely reported in the articles analyzed by this SLR. Bluetooth, which is based on the IEEE 802.15.1 standard, is characterized by high speed, low cost, and reduced coverage. In this review, evidence of its use was only found in Ref. [83]. It was one of the wireless solutions used in the evaluation of technologies, operating the tests in the 2.4 GHz band.

LoRaWAN is another specification that can be used in WSN communications infrastructure in agricultural or similar environments. It is a wide-area, low-speed, high-coveragerange, and reduced-power-consumption protocol [96]. One of its main advantages is that it works in bands below 1 GHz [97], which is advantageous in signal propagation. In Ref. [98], the author does not develop a new model and does not characterize the attenuation in the environment, but this technology was used to calculate losses in the environment using propagation models. Although the number of papers reporting the use of LoRa is small compared to Zigbee, due to its characteristics, it is possibly the most promising wireless technology in WSN-based IoT applications in scenarios that require wide coverage, as is the case of cropping systems, the reason for which is its use increased considerably between 2020 and 2021, reported in Refs. [99–102].

In the works that do not present a specific wireless communication standard and that are related to WSN planning, radiofrequency (RF) equipment (signal generators and spectrum analyzers) that are generally used in field tests were available. Although, in these circumstances, the use of the 2.4 GHz band is also predominant, and an additional set of frequencies of the radio spectrum is part of other works. In this type of study, the development of new propagation models was not a common activity, with only one report in Ref. [43], in which the selected frequency band was 433 MHz. The rest of the papers in which they used RF equipment at the time of *RSSI* measurements present attenuation characterization or model comparison activities. Within the set of frequencies evaluated in the use of such tools are the 3.5 GHz and 5.8 GHz frequencies examined in Ref. [82], the frequencies at 500 MHz, 1.2 GHz, and 2.5 GHz used in Ref. [40], while, in [41,43], they performed measurements at 870 MHz. Figure 11 presents information on the number of articles referring to each communication technology.



**Figure 11.** Number of articles that used the different communication technologies that appear in the SLR.

#### 6. Discussion

We discuss the analysis of the techniques used in different investigations. We analyze the techniques used to determine the effectiveness in terms of attenuation prediction in the models used. It is important to note that there is no uniformity in the criteria used in the evaluation of the results, and, in many cases, there is no quantitative description. In contrast, much of the work that has considered factors other than frequency and distance has focused on loss behavior at different antenna heights or the type of foliage or distribution of trees in the environment. This section presents the most used evaluation methods during SLR.

This review identified the absence of standard validation techniques that determine the effectiveness of radio wave propagation models in agricultural or forestry environments. In most of the articles, the validation of the effectiveness of the propagation models was performed based on a comparative analysis of field measurements at different distances between the transmitter and receiver nodes. Some works do not make a conversion from *RSSI* to decibel losses [18,50,60,72,90,95]. In other cases [27,40,41], the analysis was conducted from the path loss contrast, considering the depth of the vegetation.

Regarding radio spectrum, although ISM bands facilitate the deployment of WSNs in PA, unlike licensed spectrum, excessive use of the 2.4 GHz portion increases the future probability of interference due to saturation of this band. Many measurements were taken in the 2.4 GHz band, and, at times, attenuation at other frequencies was also characterized. For example, in Ref. [40], they performed tests at 500 MHz; in Ref. [43], they took data at 870 MHz, while, in Ref. [54], it was completed at 920 MHz.

More than 15 validated propagation models are identified from this SLR. However, the results show that there is a preference both for the use of certain general models not developed for agricultural, forestry, or similar environments (e.g., FSPL and Two-Ray) and the use of vegetation models (e.g., MED, COST-235, FITU-R, ITU-R, and Weissberger). Therefore, the discussion of the results has been based on the models presented in Section 5.1, without suggesting that other models should not be considered. As for the criteria used to identify the level of effectiveness of the validated models, the most frequently used techniques are the root mean square error (*RMSE*), the mean absolute percentage error (*MAPE*), and the graphical method. Of these three, the most widely used method is the *RMSE*, calculated by Equation (10) [42]:

$$RMSE = \sqrt{\frac{1}{n} \times \sum_{i=1}^{n} (x_i - \hat{x}_i)^2}.$$
 (10)

16 of 28

where  $x_i$ , y,  $\hat{x}_i$ . are measured and predicted values, respectively, while n corresponds to the number of samples taken in the experiment.

Another statistical technique used is *MAPE*, which is less susceptible to extreme values [46] and is obtained by Equation (11):

$$MAPE = \frac{1}{n_k} \sum_{i=1}^{n_k} \left| \frac{x_i - \hat{x}_i}{x_i} \right| \times 100\%.$$
(11)

where  $x_i$  is the *i*-th empirical value,  $\hat{x}_i$ . is the *i*-th theoretical value, and  $n_k$  is the number of samples. In both cases (*RMSE* and *MAPE*), the closer its value is to zero, the better the efficiency of the model [103].

Finally, many papers perform the analysis of the results from graphical means, making comparisons of distance regarding losses or *RSSI*. It is important to note that this type of analysis usually shows an underestimation or overestimation of losses without presenting quantitative information. Therefore, the results and conclusions presented could be subjective, which would reduce scientific rigor.

In the case of processes to evaluate the effectiveness of the proposed new models, the coefficient of determination, also called  $R^2$ , is usually used, which is a widely used metric when applying regressions. In this case, the closer the  $R^2$  is to a value of one, the better the correlation with the real values [104]. The coefficient of determination is calculated using Equation (12) [46]:

$$R^{2} = \frac{Explained \ Variation}{Total \ Variation} = \frac{\sum_{i=1}^{N} (x_{i} - \overline{x})^{2}}{\sum_{i=1}^{N} (\hat{x}_{i} - \overline{x})^{2}}$$
(12)

Although the statistical techniques presented in this section have been widely used in the validation of different models, each of them has characteristics that make them more useful under certain conditions. This becomes an opportunity to develop validation techniques that apply to different models and under different test conditions. Table 5 summarizes the criteria used in the delimitation of the model analysis, consistent with the information provided earlier in this SLR.

Table 5. Criteria used in SLR performance evaluation.

Canon	Measure		
Models	COST-235, ITU-R, FITU-R, Weissberger, MED, FSPL, Two-ray		
Characterization of attenuation	Loss vs. distance <i>RSSI</i> vs. distance		
Track predictions	Graphical method RMSE MAPE R <sup>2</sup>		
Prediction effectiveness	Follow measurements closely Overestimate Underestimate		

## 7. Results

The results are divided into two approaches. First, the effectiveness of the models described in Section 5.1 is examined and, in the second part, the behavior of the new models is detailed. The results were analyzed in articles with information on their effectiveness in predicting signal loss, using quantitative data shown in tables, figures, or detailed text.

As for COST-235, only Ref. [41] reports similar predictions to measurements made on different outdoor crops. In Ref. [26], the authors state that the values obtained using this model were like their field measurements. However, they presented a loss overestimation of more than 10 dB difference. The other articles show that the performance of this model was deficient, presenting even very high errors, as in Ref. [46], underestimating losses

with *MAPE* values up to 78%, or in Ref. [18], where the percentage error calculated in short distances was higher than 50%. Even when combined with FSPL and Two-Ray, the estimation of losses is far from the real measurements, as is the case presented in Ref. [53], which shows errors of 10.69% and 38.14% for the sum with each of the models.

The results obtained using the ITU-R demonstrate its inability to maintain a low level of error when describing signal loss in scenarios with the presence of vegetation, as well as the COST-235, preventing correct planning of WSN nodes. For example, in Refs. [38,67], they maintained a behavior close to the measurements, but, as the distance between the transmitter and receiver increased, they overestimated the measurements. Moreover, in Ref. [67], they used two antenna heights in two experimental scenarios; the first one places the antennas at 2.5 m and 3.4 m from the ground, presenting underestimation. The second scenario, with heights of 2.5 m and 4.4 m, fitted the measurements better. The remaining publications conclusively indicate that ITU-R is not able to characterize losses efficiently, confirming that variations in the environment and experimental conditions affect the results. Thus, in Ref. [40], the RMSE values range from 9.8 dB to 24.2 dB for different combinations of antenna heights and frequencies evaluated in mango plantations. On the other hand, under the same considerations in palm plantations, the RMSE values ranged from 4.9 dB to 18.1 dB. In this study, they combined the ITU-R model and the FSPL, achieving an average *RMSE* of 1.7 dB. For this model, there are reports of studies with high errors. For example, in the forest environment of Ref. [46], the MAPE values were 89%, and the RMSE reached 87.5 dB.

Regarding FITU-R, it only performed well in estimating losses under certain experimental configurations. In Ref. [38], the measurements and predictions are similar in leafless cherry crops, but, in the presence of foliage, there is an underestimation. In Ref. [62], this model is overestimated by reducing the height of the antennae near the canopy. In some cases, it was combined with other models not specifically developed to measure casing losses to increase its efficiency. For example, in Ref. [18], when combined with FSPL, the percentage error was less than 5% when the distance from the transmitter to the receiver was equal to or greater than 9.8 meters, but, at short distances, the error is high, reaching 35.22% at 2.6 meters. The rest of the articles analyzed show that the FITU-R is inadequate for predicting losses in the scenarios considered, although they do not report excessively high errors, unlike what occurred when other models (COST-235 and ITU-R) were used. In some scenarios, it presents underestimation when using the model [53], while, in Ref. [60], its prediction was above the measurements.

Regarding Weissberger, there is no information on the low levels of error in his estimation that establishes signal loss. Even with adjustments, as is the case presented in Ref. [42], the  $R^2$  only reaches a value of 0.67. The results of most papers show a tendency for this model to underestimate losses. Only in Ref. [18] were the FSPL and Two-Ray models combined, showing an overestimation in the measurements made inside a tomato greenhouse and an antenna height of 0.5 meters at its nodes. In Ref. [46], they recorded the highest error of this model, reaching a *MAPE* value of 39%.

Finally, concerning the vegetation models, the MED performed well in the application scenarios, although this does not indicate that it is suitable for sizing WSNs in agricultural or similar environments. In Ref. [40], they performed an adjustment process to improve its performance, obtaining *RMSE* values of 10.6 dB at 2.2 m and 2.57 dB at a measurement height of 2.6 meters. However, the band used was the 400 MHz band, which has much lower losses in contrast to other more commonly used bands, for example, the 2.4 GHz band.

The FSPL and Two-Ray models are widely used, either in comparison or in combination with other models. Regarding FSPL, all the articles report results that indicate significant differences when compared with measurements made in different study settings. In addition, there is no tendency to overestimate or underestimate. In Refs. [43,46,53], the solutions given by the model at each distance were below the obtained measurements. In contrast, in Ref. [56], he estimated above the real data. In all the articles analyzed, the change in the conditions in the experiment affected the results. For example, in Ref. [18], there was an underestimation when the antennas were located at 0.5 m, while, when increasing to 1.5 m, the behavior was of overestimation. Possibly due to its design characteristics, it presented high differences concerning the measurements. In Ref. [105], the minimum *MAPE* obtained was 30% at short distances, while Ref. [53] recorded values up to 82.44% using the same statistical analysis.

As for Two-Ray, there are also no case reports indicating an approximation to the empirical data. Its use generated overly high errors, especially in Ref. [18], where the percentage error was over 90% at any distance between the transmitter and receiver at an antenna height of 1.5 m, although, when reduced to 0.5 m, it decreased considerably to a maximum of 26.25%.

The compilation of results presented in the different works considered in this SLR demonstrates the low efficiency of traditional vegetation models when used as tools for loss estimation between the transmitter and receiver of a WSN. In addition, the lack of standardization in terms of the validation of the results and the limited information provided by some authors prevent a more detailed analysis of the results obtained in some experiments. However, the information gathered allows inferring that the vegetation models historically used in the characterization of attenuation in PA environments or with the presence of vegetation are not the most appropriate when used in the sizing of WSNs in these environments. Table 6 presents a summary of the references selected in the evaluation, with the results obtained in each case from the previously validated models.

As for the research aimed at developing new models considered in this review, the results are acceptable from a statistical point of view. In most cases, the criterion used was the  $R^2$ , although there are reports of analysis of results based on *RMSE* and *MAPE*.

In three of the articles reviewed, the low estimates of the nontraditional models developed are reported. In Ref. [60], only 65.82% of the loss variations could be explained by the proposed regression model, a result that corresponds to the case where the node antenna is located at a height of 3.5 m since, at a height of 7 m, the  $R^2$  was 0.94. Ref. [50] also reports that the proposed development presents an acceptable behavior in the estimation of losses in a rice field during its different growth stages but does not present quantitative data. Finally, in Ref. [38], they indicate that the proposed model provides good agreement near the tree canopy but underestimates the attenuation at mid-height. Moreover, in this case, the error increases with the presence of leaves and with increasing distance. In the rest of the articles, the criteria used in the evaluation of the proposed models yielded favorable values, so it can be affirmed that they are capable of characterizing, from a statistical point of view, the attenuation in each of the environments studied. For example, in Refs. [78,105,106], the values of  $R^2$  were 0.95, 0.98, and 0.98, respectively.

The *RMSE* was also used to validate the results of the new models. In Ref. [43], they used it in the validation phase of the effectiveness of their proposal concerning field measurements, obtaining values of 1.6 dB in the 870 MHz band and 1.5 dB in the 2.4 GHz band. In Ref. [62], they reported that the calculated *RMSE* was less than 1.5 dB for all the antenna heights and crop conditions considered.

As for the relative error, only Ref. [48] reports its use as a statistic for the validation of the results. In this case, the proposed model had a difference of less than 2% concerning attenuation variations.

The results associated with the development of new models contrast with those obtained with traditional vegetation models. In this case, the models obtained from the measurements and the considerations of the specific environment in which the WSN is planned to be installed allow modeling losses with much lower error levels. However, the multiple variations of crops and associated environments represent a challenge for future researchers since it has not been possible to obtain a standard model that can be applied in all scenarios with good results and that avoids the characterization of each crop or wooded area where a WSN is intended to be installed. Table 7 presents a summary of the new models and their results.

**Table 6.** Summary of the results of the evaluation of the previously validated models.

Reference	Models	Environment	Technology	Summary of Results
[16]	FSPL, Two-Ray	Grass	RF equipment	- Two-Ray and FSPL do not adequately represent measurements.
[12]	COST-235, ITU-R, Weissberger	Agricultural fields	Zigbee	<ul> <li>COST-235 was the largest overvaluation of losses in all growth stages of the crop.</li> <li>Weissberger was the one with the greatest underestimation of losses in all stages of the crop.</li> </ul>
[27]	COST-235, FSPL, Weissberger	Agricultural fields	RF equipment	<ul> <li>FSPL, Weissberger, and ITU-R show similar trends and differ from the observed path loss values.</li> <li>COST 235 features higher precision.</li> </ul>
[38]	ITU-R	Agricultural fields	Zigbee	<ul> <li>FITU-R was the most accurate in a setting with no leaves on the trees.</li> <li>FSPL presented the highest difference concerning the measures.</li> </ul>
[40]	FITU-R, ITU-R, Weissberger	Agricultural fields	RF equipment	<ul> <li>FITU-R was more accurate at frequencies below 2 GHz.</li> <li>COST235 was more accurate at frequencies above 2 GHz.</li> </ul>
[39]	FSPL, MED	Agricultural fields	Zigbee	<ul> <li>FSPL, adjusted to the measurements, was the most accurate with grass heights of 15 cm (<i>RMSE</i> of 4.5) and 1 m (1.62).</li> <li>MED, adjusted to measurements, was the least accurate with grass heights of 15 cm (<i>RMSE</i> of 4.65) and 1 m (1.9).</li> </ul>
[42]	ITU-R, Weissberger	Agricultural fields	Zigbee	- Weissberger was the most accurate with an R2 of 0.62 when adjusted.
[43]	COST-235, FITU-R, FSPL, ITU-R	Agricultural fields	RF equipment	<ul> <li>FITU-R was more accurate at 2.4 GHz.</li> <li>ITU-R and FITU-R show similar trends at 870 MHz.</li> <li>FSPL was the one with the greatest difference concerning the measurements.</li> </ul>
[44]	COST-235, FITU-R, Weissberger	Agricultural fields	Zigbee	<ul> <li>FITU-R, in all the combinations of the experiment, presented the highest precision (<i>RMSE</i> between 3.8 and 13.6).</li> <li>COST-235, in all the combinations of the experiment, presented the highest overvaluation (<i>RMSE</i> between 8.0 and 26.8).</li> </ul>

Tabla	6	Cont
Idvie	υ.	Com.

Reference	e Models	Environment	Technology	Summary of Results
[18]	COST-235, FSPL, ITU-R, Two-Ray, Weissberger	Greenhouse	Zigbee	<ul> <li>Two-Ray presented was more accurate with node antenna height of 0.5 m (% error between 0.65% and 26.25%).</li> <li>FSPL presented the greatest difference with node antenna height of 0.5 m (% error between 4.52% and 42.84%).</li> <li>ITU-R added with Two-Ray and Weissberger added with Two-Ray were the most accurate with a node antenna height of 1.5 m.</li> </ul>
[46]	COST-235, FSPL, ITU-R, Two-Ray, Weissberger	Forest or jungle	Zigbee	<ul> <li>Two-Ray was more accurate (<i>MAPE</i> 25%).</li> <li>ITU-R was the one with the greatest difference concerning the measurements (<i>MAPE</i> of 89%).</li> </ul>
[48]	FSPL, Two-Ray	Agricultural fields	Zigbee	<ul> <li>FSPL presented a low performance in all stages of the culture (<i>RMSE</i> between 10.33 dB and 18.19 dB).</li> <li>Two-Ray showed a decrease in <i>RMSE</i> when increasing the height of the node antenna.</li> </ul>
[53]	COST-235, FITU-R, FSPL, ITU-R, Two-Ray, Weissberger	Greenhouse	Zigbee	<ul> <li>COST-235 added with FSPL was the most accurate (<i>MAPE</i> of 10.69%).</li> <li>Weissberger added with Two-Ray presented the greatest difference concerning the measurements.</li> </ul>
[56]	FSPL	Forest or jungle	WiFi	- FSPL presented high overvaluation of the measurements.
[58]	COST-235, FITU-R, FSPL, Weissberger	Forest or jungle	RF equipment	- All models underestimated the measurements ( <i>RMSE</i> up to 68 dB at 25 m).
[60]	COST-235, FITU-R	Forest or jungle	WiFi	- FITU-R and COST-235 presented high overvaluation of the measurements.
[82]	FSPL, ITU-R	Forest or jungle	RF equipment	<ul> <li>ITU-R overestimates at distances less than 20 m and underestimates at distances greater than 20 m.</li> <li>FSPL overestimates measurements.</li> </ul>
[67]	COST-235, FITU-R, ITU-R, MED	Agricultural fields	Zigbee	<ul> <li>MED was more accurate with antenna heights of 2.0 m and 2.3 m.</li> <li>ITU-R had good prediction in short distances (up to 54 m) with an antenna height of 2.3 m.</li> <li>FITU-R improved performance as distance increased with 2.0 m and 2.3 m antenna height.</li> <li>COST-235 overestimated measurements in all cases.</li> </ul>

Reference	Models	Environment	Technology	Summary of Results
[104]	FSPL, Two-Ray	Forest and grass	Zigbee	<ul> <li>Two-Ray was the most accurate (average <i>RMSE</i> of 19.8).</li> <li>FSPL presented the highest difference concerning the measurements (mean <i>RMSE</i> of 34.37).</li> </ul>
[106]	FSPL, ITU-R, Two-Ray	Agricultural fields	Zigbee	<ul> <li>Without adjusting, Two-Ray was more accurate with an antenna height of 0.5 m (average <i>RMSE</i> of 10.33 dB).</li> <li>Without adjusting, FSPL was more accurate with antenna heights of 1.0 m and 1.5 m (average <i>RMSE</i> of 10.22 dB and 8.26 dB, respectively).</li> <li>Without adjusting, ITU-R had the highest difference concerning the measurements in all scenarios (<i>RMSE</i> greater than 60 dB).</li> <li>By adjusting Two-Ray, it was the most accurate in all scenarios (average <i>RMSE</i> less than 2.5 dB).</li> <li>When adjusting, it presented a greater difference with the measurements (average <i>RMSE</i> between 1.93 dB to 2.06 dB).</li> </ul>

Reference	Environment	Technology	Summary of Results
[43]	Forest or jungle	RF equipment	<ul> <li><i>RMSE</i> = 1.6 dB at 870 MHz.</li> <li><i>RMSE</i> = 1.5 dB at 2.4 GHz.</li> </ul>
[44]	Agricultural field	Zigbee	<ul> <li>Average <i>RMSE</i> of 4.48 in all combinations of the experiment.</li> <li><i>R</i><sup>2</sup> between 0.87 and 0.9 with logarithmic adjustment.</li> </ul>
[48]	Agricultural field	Zigbee	- Relative error < 2%
[60]	Forest or jungle	WiFi	<ul> <li>R<sup>2</sup> = 0.66 with antenna height of 3.5 m.</li> <li>R<sup>2</sup> = 0.94 with antenna height of 7 m.</li> </ul>
[38]	Agricultural field	Zigbee	<ul> <li>Good match near the treetops.</li> <li>Underestimation at a medium height.</li> <li>The error increased with the presence of leaves and with increasing distance.</li> </ul>
[18]	Greenhouse	Zigbee	<ul> <li>% average error = 2.76% with a minimum of 0.06% and a maximum of 8.81% at distances of 2.6 m and 13.4 m, respectively, with antenna height of 0.5 m.</li> <li>% average error = 1.57% with a minimum of 0.26% and a maximum of 3.01% at distances of 6.2 m and 4.4 m, respectively, with antenna height of 1.5 m.</li> </ul>
[50]	Agricultural field	Zigbee	- Acceptable results.
[62]	Forest or jungle	RF equipment	<ul> <li><i>RMSE</i> between 0.97 to 1.49 with antenna height of 2 m.</li> <li><i>RMSE</i> 0.19 to 1.33 with 3 m antenna height.</li> </ul>
[77]	Grass	Zigbee	- $R^2 = 0.95$ in short grass. - $R^2 = 0.92$ in tall grass.
[104]	Forestry and turf	Zigbee	<ul> <li>R<sup>2</sup> = 0.99 in grass.</li> <li>R<sup>2</sup> = 1 in a sparse tree environment.</li> </ul>
[105]	Forest or jungle	Zigbee	$ R^2 = 0.98$
[106]	Agricultural field	RF equipment	<ul> <li>R<sup>2</sup> = 0.98 under (LoS) conditions.</li> <li>R<sup>2</sup> = 0.98 under non-line-of-sight (NLoS) conditions.</li> </ul>

Table 7. Summary of evaluation results of new models.

## 8. Conclusions

The results of this SLR show that, although there is a large set of radio wave propagation models validated in vegetated environments, the characterization of attenuation in agricultural, or similar, environments is an area of engineering that requires further research. In addition, this SLR identifies the models most used in planning the sizing of WSNs in crops, as well as the scenarios in which most research has been conducted, as well as trends in the use of wireless technologies required for node connectivity. The criteria and values that determine the validity of a model and/or its effectiveness were also identified.

The number of articles in this research area had been increasing from 2015 to 2019, decreasing from 2020, probably a product of the SARS-CoV-2 pandemic, which forced the suspension of activities in different research areas. Although the number of articles found in publications not categorized as proceedings is high (50%), it is also important to note

that 26% of the papers consulted in this SLR are found in journals or proceedings of events classified as Q1, which demonstrates the relevance of the research area for journals that address this type of topic.

Six countries account for most of the publications generated (China, India, Japan, Malaysia, Spain, and the United States). The remaining articles come from 18 other nations, demonstrating that there is a diversity of institutions around the world researching this topic.

The technical analysis allows us to identify that, of the validated and recognized models, the most widely used are of the empirical type, highlighting the use of the MED and its derivatives. However, the results of *RMSE* above 10 dB and *MAPE* above 10% in most investigations show that vegetation models are inefficient in estimating radio signal loss in foliage when applied in scenarios other than those considered in their development. Therefore, adjustments are needed to increase the accuracy under specific experimental conditions and in the configuration of the communication system elements.

Several papers consider non-specific models for vegetated environments (e.g., FSPL and Two-Ray). However, these are generally used as an aid to calculate attenuation over the entire communication link and not only in vegetation. Like vegetation models, they lead to high errors. The results reported on the use of vegetation models and their effectiveness in estimating attenuation in the scenarios studied indicate that none can be rated as the most accurate in predicting signal losses in environments with the presence of bushes, plants, grass, or similar. All the vegetation models considered showed variations in their accuracy in the cases in which they were applied, with no definite trend of estimation above or below the real loss values. The poor effectiveness of vegetation models demonstrates the need for more comprehensive developments that include factors other than distance and frequency, e.g., antenna height, obstacle dimensions, and shrub spacing.

As for the models considered new, most of them adequately fitted the attenuation curve in the scenarios in which they were developed, and, in the conditions proposed in different experiments, reached  $R^2$  values higher than 0.9 in most of the investigations.

Regarding the scenarios considered in the characterization of attenuation and the evaluation or development of new models, most of the studies focus on outdoor agricultural environments and greenhouses. In these, since there is generally a uniform layout and one type of shrub predominates in the fields, modeling the effects of vegetation on the signal in the wireless channel is facilitated. As for environments dominated by grasses or pastures (common in livestock environments), they present favorable conditions for radio wave propagation. However, the results show that, in most cases, it is necessary to adjust the vegetation models to reduce error levels. As for forest environments, which, in many cases, have similarities with some types of crops, they represent a great challenge in the estimation of signal loss because the uniform positioning of trees and plants does not predominate. In addition, it is normal that there are different species of vegetation, and each of them has its geometrical characteristics.

Regarding the wireless technologies used in WSNs for agricultural applications, the results show that, from 2011 to 2021, there is a preference for the use of Zigbee. However, as of 2019, there is a growing trend in the use of LoRa. Its coverage, power consumption, and cost characteristics position it as one of the wireless technologies with the highest future demand in WSN and IoT applications.

### 9. Future Works

Future developments of new models should further study the effects of plant geometry, composition, and layout on radio signal loss. In the case of greenhouse crops, it is desirable to study how the infrastructure that houses them influences signal attenuation. For the study of radio wave propagation in agricultural environments or with the presence of bushes or grass, it is advisable to use techniques that have not been used in studies of radio wave propagation in these scenarios, such as machine learning. In addition, this SLR identifies the lack of standard validation techniques to adequately evaluate and compare

propagation models centered on PA scenarios, opening the possibility of lines of research in this area. It is also suggested to analyze those technologies that have not been widely documented and that have interesting features, such as large coverage, e.g., Sigfox.

Regarding the application context, studies are proposed focused on crops of great importance for food sustainability, such as cassava, banana, or soybean, and ecosystems located in marine–coastal transition zones, such as mangrove forests, among others.

Author Contributions: Conceptualization, A.B.-U. and E.D.I.H.-F.; methodology, E.D.I.H.-F.; validation, P.P.A.-C. and A.P.Q.-L.; formal analysis, P.P.A.-C. and A.P.Q.-L.; investigation, A.B.-U.; data curation, P.P.A.-C. and A.P.Q.-L.; writing—original draft preparation, E.D.I.H.-F.; writing—review and editing, A.B.-U. and H.S.-M.; visualization, P.P.A.-C. and A.P.Q.-L.; supervision, E.D.I.H.-F.; project administration, E.D.I.H.-F.; funding acquisition, A.B.-U. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Conflicts of Interest: The authors declare no conflict of interest.

#### References

- 1. Sharma, R.P.; Ramesh, D.; Pal, P.; Tripathi, S.; Kumar, C. Crop Pest Prediction. IEEE Internet Things J. 2022, 9, 3037–3045. [CrossRef]
- Al-Qurabat, A.K.M. A Lightweight Huffman-based Differential Encoding Lossless Compression Technique in IoT for Smart Agriculture. Int. J. Comput. Digit. Syst. 2022, 11, 117–127. [CrossRef]
- Sabri, N.; Mohammed, S.S.; Fouad, S.; Syed, A.A.; Al-Dhief, F.T.; Raheemah, A. Investigation of Empirical Wave Propagation Models in Precision Agriculture. *MATEC Web Conf.* 2018, 150, 06020. [CrossRef]
- 4. Khairunnniza-Bejo, S.; Ramli, N.H.; Muharam, F.M. Wireless sensor network (WSN) applications in plantation canopy areas: A review. *Asian J. Sci. Res.* 2018, *11*, 151–161. [CrossRef]
- 5. Thakur, D.; Kumar, Y.; Kumar, A.; Singh, P.K. *Applicability of Wireless Sensor Networks in Precision Agriculture: A Review*; Springer: Berlin/Heidelberg, Germany, 2019; ISBN 0123456789.
- Pal, P.; Sharma, R.P.; Tripathi, S.; Kumar, C.; Ramesh, D. 2.4 GHz RF Received Signal Strength Based Node Separation in WSN Monitoring Infrastructure for Millet and Rice Vegetation. *IEEE Sens. J.* 2021, 21, 18298–18306. [CrossRef]
- Barrios-Ulloa, A.; Cama-PInto, D.; Mardini-Bovea, J.; Díaz-Martínez, J.; Cama-Pinto, A. Projections of IoT Applications in Colombia Using 5G Wireless Networks. Sensors 2021, 21, 7167. [CrossRef] [PubMed]
- Cama-Pinto, D.; Damas, M.; Holgado-Terriza, J.A.; Gómez-Mula, F.; Calderín-Curtidor, A.C.; Martínez-Lao, J.A.; Cama-Pinto, A. 5G Mobile Phone Network Introduction in Colombia. *Electronics* 2021, 10, 922. [CrossRef]
- 9. Dogan, H. A new empirical propagation model depending on volumetric density in citrus orchards for wireless sensornetwork applications at sub-6 GHz frequency region. *Int. J. RF Microw. Comput. Eng.* **2021**, *31*, e22778. [CrossRef]
- 10. Gabriel, P.-E.; Butt, S.A.; Francisco, E.-O.; Alejandro, C.-P.; Maleh, Y. Performance analysis of 6LoWPAN protocol for a flood monitoring system. *EURASIP J. Wirel. Commun. Netw.* **2022**, 2022, 16. [CrossRef]
- Cama-Pinto, A.; Piñeres-Espitia, G.; Comas-González, Z.; Vélez-Zapata, J.; Gómez-Mula, F. Design of a monitoring network of meteorological variables related to tornadoes in Barranquilla-Colombia and its metropolitan area. *Ingeniare* 2017, 25, 585–598. [CrossRef]
- 12. Wu, H.; Zhang, L.; Miao, Y. The Propagation Characteristics of Radio Frequency Signals for Wireless Sensor Networks in Large-Scale Farmland. *Wirel. Pers. Commun.* 2017, *95*, 3653–3670. [CrossRef]
- Anusha, V.S.; Nithya, G.K.; Rao, S.N. A comprehensive survey of electromagnetic propagation models. In Proceedings of the 2017 International Conference on Communication and Signal Processing (ICCSP), Chennai, India, 6–8 April 2017; pp. 1457–1462.
- 14. Ganev, Z. Log-normal shadowing model for outdoor propagation between sensor nodes. In Proceedings of the 2018 20th International Symposium on Electrical Apparatus and Technologies (SIELA), Bourgas, Bulgaria, 3–6 June 2018; pp. 9–12.
- 15. Kurt, S.; Tavli, B. Path-Loss Modeling for Wireless Sensor Networks: A review of models and comparative evaluations. *IEEE Antennas Propag. Mag.* 2017, 59, 18–37. [CrossRef]
- Tang, W.; Ma, X.; Wei, J.; Wang, Z. Measurement and analysis of near-ground propagation models under different terrains for wireless sensor networks. *Sensors* 2019, 19, 1901. [CrossRef]
- 17. Vougioukas, S.; Anastassiu, H.T.; Regen, C.; Zude, M. Influence of foliage on radio path losses (PLs) for Wireless Sensor Network (WSN) planning in orchards. *Biosyst. Eng.* **2013**, *114*, 454–465. [CrossRef]
- 18. Cama-Pinto, D.; Damas, M.; Holgado-Terriza, J.A.; Gómez-Mula, F.; Cama-Pinto, A. Path loss determination using linear and cubic regression inside a classic tomato greenhouse. *Int. J. Environ. Res. Public Health* **2019**, *16*, 1744. [CrossRef]
- 19. Rappaport, T.S. Wireless Communications: Principles and Practice, 2nd ed.; Prentice Hall: Hoboken, NJ, USA, 2020.
- De Sales Bezerra, T.; De Sousa, J.A.R.; Da Silva Eleuterio, S.A.; Rocha, J.S. Accuracy of propagation models to power prediction in WSN ZigBee applied in outdoor environment. In Proceedings of the 6th Argentine Conference on Embedded Systems (CASE), Buenos Aires, Argentina, 12–14 August 2015; pp. 19–24.

- Barrios, A.; Arjona, R.; Álvarez, R. Comparison of Radio Propagation Models in the Suburban Area of the City of Barranquilla. *Rev. Colomb. Tecnol. Av.* 2018, 2, 78–85. [CrossRef]
- Shoewu, O.O.; Akinyemi, L.A.; Oborkhale, L. Towards Developing Path loss Models for Dryland and Wetland Environments. In Proceedings of the IEEE AFRICON Conference, Accra, Ghana, 25–27 September 2019. [CrossRef]
- Sevgi, L. Groundwave modeling and simulation strategies and path loss prediction virtual tools. *IEEE Trans. Antennas Propag.* 2007, 55, 1591–1598. [CrossRef]
- 24. Phillips, C.; Sicker, D.; Grunwald, D. A Survey of Wireless Path Loss Prediction and Coverage Mapping Methods. *IEEE Commun. Surv. Tutor.* **2013**, *15*, 255–270. [CrossRef]
- Kamarudin, L.M.; Ahmad, R.B.; Ong, B.L.; Malek, F.; Zakaria, A.; Arif, M.A.M. Review and modeling of vegetation propagation model for wireless sensor networks using OMNeT++. In Proceedings of the 2nd International Conference on Network Applications, Protocols and Services (NETAPPS), Alor Setar, Malaysia, 22–23 September 2010; pp. 78–83.
- Rama Rao, T.; Balachander, D.; Tiwari, N. RF Propagation Measurements in Forest & Plantation Environments for Wireless Sensor Networks. In Proceedings of the 2012 IEEE International Conference on Communication Systems (ICCS), Chennai, India, 19–21 April 2012; pp. 194–198.
- Sabri, N.; Aljunid, S.A.; Salim, M.S.; Fouad, S. Wireless Sensor Network Wave Propagation in Vegetation. In Recent Trends in Physics of Material Science and Technology; Springer: Singapore, 2015; Volume 204, pp. 283–298. [CrossRef]
- Kumar, S.A.; Ilango, P. The Impact of Wireless Sensor Network in the Field of Precision Agriculture: A Review. Wirel. Pers. Commun. 2018, 98, 685–698. [CrossRef]
- Kochhar, A.; Kumar, N. Wireless sensor networks for greenhouses: An end-to-end review. Comput. Electron. Agric. 2019, 163, 104877. [CrossRef]
- Cisternas, I.; Velásquez, I.; Caro, A.; Rodríguez, A. Systematic literature review of implementations of precision agriculture. *Comput. Electron. Agric.* 2020, 176, 105626. [CrossRef]
- 31. Ojha, T.; Misra, S.; Raghuwanshi, N.S. Wireless sensor networks for agriculture: The state-of-the-art in practice and future challenges. *Comput. Electron. Agric.* 2015, *118*, 66–84. [CrossRef]
- 32. Velásquez, J.D. A short guide to writing systematic literature reviews. Part 1. DYNA 2014, 81, 9–10. [CrossRef]
- 33. Velásquez, J.D. A short guide to writing systematic literature reviews. Part 2. DYNA 2015, 82, 9–12. [CrossRef]
- 34. Velásquez, J.D. A short guide to writing systematic literature reviews. Part 3. DYNA 2015, 82, 9–12. [CrossRef]
- 35. Suter, G.W. Review papers are important and worth writing. Environ. Toxicol. Chem. 2013, 32, 1929–1930. [CrossRef]
- De-La-Hoz-Franco, E.; Ariza-Colpas, P.; Quero, J.M.; Espinilla, M. Sensor-based datasets for human activity recognition—A systematic review of literature. *IEEE Access* 2018, 6, 59192–59210. [CrossRef]
- 37. ITU-R. ITU-R. Recommendation P.833-7 Attenuation in Vegetation; ITU-R: Geneva, Switzerland, 2012; Volume 7.
- Anastassiu, H.T.; Vougioukas, S.; Fronimos, T.; Regen, C.; Petrou, L.; Zude, M.; Käthner, J. A computational model for path loss in wireless sensor networks in orchard environments. *Sensors* 2014, 14, 5118–5135. [CrossRef] [PubMed]
- Ndzi, D.L.; Harun, A.; Ramli, F.M.; Kamarudin, M.L.; Zakaria, A.; Shakaff, A.Y.M.; Jaafar, M.N.; Zhou, S.; Farook, R.S. Wireless sensor network coverage measurement and planning in mixed crop farming. *Comput. Electron. Agric.* 2014, 105, 83–94. [CrossRef]
- Ndzi, D.L.; Kamarudin, L.M.; Mohammad, E.A.A.; Zakaria, A.; Ahmad, R.B.; Fareq, M.M.A.; Shakaff, A.Y.M.; Jafaar, M.N. Vegetation attenuation measurements and modeling in plantations for wireless sensor network planning. *Prog. Electromagn. Res.* B 2011, 36, 283–301. [CrossRef]
- Balachander, D.; Rao, T.R.; Mahesh, G. RF propagation investigations in agricultural fields and gardens for wireless sensor communications. In Proceedings of the 2013 IEEE Conference on Information & Communication Technologies, Thuckalay, India, 11–12 April 2013; pp. 755–759.
- Correia Pinheiro, F.; Sampaio De Alencar, M.; Araújo Lopes, W.T.; Soares De Assis, M.; Gonçalves Leal, B. Propagation analysis for wireless sensor networks applied to viticulture. *Int. J. Antennas Propag.* 2017, 2017. [CrossRef]
- 43. Azevedo, J.A.; Santos, F.E. A model to estimate the path loss in areas with foliage of trees. *AEU Int. J. Electron. Commun.* 2017, 71, 157–161. [CrossRef]
- 44. Guo, X.-M.; Yang, X.-T.; Chen, M.-X.; Li, M.; Wang, Y.-A. A model with leaf area index and apple size parameters for 2.4 GHz radio propagation in apple orchards. *Precis. Agric.* 2015, *16*, 180–200. [CrossRef]
- Wu, H.; Zhu, H.; Han, X.; Xu, W. Layout optimization for greenhouse WSN based on path loss analysis. *Comput. Syst. Sci. Eng.* 2021, 37, 89–104. [CrossRef]
- Olasupo, T.O.; Otero, C.E. The Impacts of Node Orientation on Radio Propagation Models for Airborne-Deployed Sensor Networks in Large-Scale Tree Vegetation Terrains. *IEEE Trans. Syst. Man Cybern. Syst.* 2020, 50, 256–269. [CrossRef]
- 47. Cama-Pinto, D.; Damas, M.; Holgado-Terriza, J.A.; Arrabal-Campos, F.M.; Gómez-Mula, F.; Lao, J.A.M.; Cama-Pinto, A. Empirical model of radio wave propagation in the presence of vegetation inside greenhouses using regularized regressions. *Sensors* **2020**, 20, 6621. [CrossRef]
- Gao, Z.; Li, W.; Zhu, Y.; Tian, Y.; Pang, F.; Cao, W.; Ni, J. Wireless channel propagation characteristics and modeling research in rice field sensor networks. *Sensors* 2018, 18, 3116. [CrossRef] [PubMed]
- 49. Devarajan, N.; Gupta, S.H. Implementation and Analysis of Different Path Loss Models for Cooperative Communication. In *Smart Innovations in Communication and Computational Sciences*; Springer: Singapore, 2019; pp. 227–236. [CrossRef]

- Hamasaki, T. Propagation Characteristics of A 2.4GHz Wireless Sensor Module with a Pattern Antenna in Forestry and Agriculture Field. In Proceedings of the 2019 IEEE International Symposium on Radio-Frequency Integration Technology (RFIT), Nanjing, China, 28–30 August 2019; pp. 1–3.
- Alsayyari, A.; Aldosary, A. Path Loss Results for Wireless Sensor Network Deployment in a Long Grass Environment. In Proceedings of the 2018 IEEE Conference on Wireless Sensors (ICWiSe), Istanbul, Turkey, 18–20 June 2019; pp. 50–55.
- AlSayyari, A.; Kostanic, I.; Otero, C.E. An empirical path loss model for Wireless Sensor Network Deployment in a Sand Terrain Environment. In Proceedings of the 2014 IEEE World Forum on Internet of Things (WF-IoT), Seoul, Korea, 6–8 March 2014; pp. 218–223.
- 53. Raheemah, A.; Sabri, N.; Salim, M.S.; Ehkan, P.; Ahmad, R.B. New empirical path loss model for wireless sensor networks in mango greenhouses. *Comput. Electron. Agric.* 2016, 127, 553–560. [CrossRef]
- Daisuke, K.; Tatsuki, T.; Toshihiko, H. Vegetation Effect in Paddy Field for a Wireless Sensor Network. In Proceedings of the 2018 USNC-URSI Radio Science Meeting (Joint with AP-S Symposium), Boston, MA, USA, 8–13 July 2018; pp. 113–114.
- 55. Klaina, H.; Picallo, I.; Iturri, P.L.; Azpilicueta, L.; Celaya-echarri, M.; Aghzout, O. Deterministic Radio Channel Characterization for Near-Ground Wireless Sensor Networks Deployment Optimization in Smart Agriculture. In Proceedings of the 2020 14th European Conference on Antennas and Propagation (EuCAP), Copenhagen, Denmark, 15–20 March 2020; pp. 3–7.
- Hakim, G.P.N.; Alaydrus, M.; Bahaweres, R.B. Empirical approach of ad hoc path loss propagation model in realistic forest environments. In Proceedings of the 2016 International Conference on Radar, Antenna, Microwave, Electronics and Telecommunications (ICRAMET), Jakarta, Indonesia, 3–5 October 2016; pp. 139–143. [CrossRef]
- Cama-Pinto, A.; Piñeres-Espitia, G.; Caicedo-Ortiz, J.; Ramírez-Cerpa, E.; Betancur-Agudelo, L.; Gómez-Mula, F. Received strength signal intensity performance analysis in wireless sensor network using Arduino platform and XBee wireless modules. *Int. J. Distrib. Sens. Netw.* 2017, 13, 2–8. [CrossRef]
- 58. Shutimarrungson, N.; Wuttidittachotti, P. Realistic propagation effects on wireless sensor networks for landslide management. *EURASIP J. Wirel. Commun. Netw.* **2019**, *94*. [CrossRef]
- 59. Artemenko, O.; Rubina, A.; Nayak, A.H.; Baptist, S.; Mitschele-thiel, A. Evaluation of different signal propagation models for a mixed indoor-outdoor scenario using empirical data. *EAI Endorsed Trans. Mob. Commun. Appl.* **2016**, *2*, 94. [CrossRef]
- Stewart, J.; Stewart, R.; Kennedy, S. Internet of Things—Propagation Modelling for Precision Agriculture Applications. In Proceedings of the IEEE International Conference Image Information Processing, Chicago, IL, USA, 26–28 April 2017; pp. 1–8.
- Jawad, H.M.; Jawad, A.M.; Nordin, R.; Gharghan, S.K.; Abdullah, N.F.; Ismail, M.; Abu-Alshaeer, M.J. Accurate Empirical Path-Loss Model Based on Particle Swarm Optimization for Wireless Sensor Networks in Smart Agriculture. *IEEE Sens. J.* 2020, 20, 552–561. [CrossRef]
- Picallo, I.; Klaina, H.; Lopez-Iturri, P.; Aguirre, E.; Celaya-Echarri, M.; Azpilicueta, L.; Eguizábal, A.; Falcone, F.; Alejos, A. A radio channel model for D2D communications blocked by single trees in forest environments. *Sensors* 2019, 19, 4606. [CrossRef] [PubMed]
- 63. Chen, Y.; Chamadiya, B. Propagation model for vehicle to vehicle LOS communication in foliage scenario. In Proceedings of the 2014 International Conference on Connected Vehicles and Expo, ICCVE, Vienna, Austria, 3–7 November 2014; pp. 1120–1125.
- 64. Rao, T.R.; Balachander, D.; Tiwari, N.; Prasad, M.V.S.N. Ultra-high frequency near-ground short-range propagation measurements in forest and plantation environments for wireless sensor networks. *IET Wirel. Sens. Syst.* **2013**, *3*, 80–84. [CrossRef]
- Rahim, H.M.; Leow, C.Y.; Rahman, T.A. Millimeter wave propagation through foliage: Comparison of models. In Proceedings of the 2015 IEEE 12th Malaysia International Conference on Communications, MICC, Kuching, Malaysia, 23–25 November 2015; pp. 236–240.
- Zhang, J.; Li, C.; Yu, J.; Li, F.; Chang, F.; Chen, Y.; Chen, W. Path Loss Analysis and Model Selection in Forest City Environment at 5.9 GHz. In Proceedings of the 2nd IEEE Advanced Information Management, Communicates, Electronic and Automation Control Conference, IMCEC, Xi'an, China, 25–27 May 2018; pp. 516–520.
- 67. Rao, Y.; Jiang, Z.H.; Lazarovitch, N. Investigating signal propagation and strength distribution characteristics of wireless sensor networks in date palm orchards. *Comput. Electron. Agric.* 2016, 124, 107–120. [CrossRef]
- Castellanos, G.; Teuta, G. Path loss model in amazonian border region for VHF and UHF television bands. In Proceedings of the 2017 IEEE-APS Topical Conference on Antennas and Propagation in Wireless Communications, APWC, Verona, Italy, 11–15 September 2017; pp. 137–140.
- Oda, E.; Kawauchi, K.; Hamasaki, T. Support Application for Configuring Optimal Relay Nodes in Wireless Sensor Networks. In Proceedings of the 2021 IEEE Topical Conference on Wireless Sensors and Sensor Networks, WiSNeT, Diego, CA, USA, 17–20 January 2021; pp. 19–22.
- Kamarudin, L.M.; Ahmad, R.B.; Ndzi, D.; Zakaria, A.; Ong, B.L.; Kamarudin, K.; Harun, A.; Mamduh, S.M. Modeling and simulation of WSNs for agriculture applications using dynamic transmit power control algorithm. In Proceedings of the 3rd Inernational Conference on Intelligent Systems Modelling and Simulation, Kota Kinabalu, Malaysia, 8–10 February 2012; pp. 616–621. [CrossRef]
- 71. Myagmardulam, B.; Tadachika, N.; Takahashi, K.; Miura, R.; Ono, F.; Kagawa, T. Path Loss Prediction Model Development in a Mountainous Forest Environment. *IEEE Open J. Commun. Soc.* **2021**, *2*, 2494–2501. [CrossRef]

- Yamaoka, Y.; Hamasaki, T.; Kuramoto, D. 2.4 GHz RF Propagation Measurements and Modeling in a Paddy Field for a Wireless Sensor Network. In Proceedings of the 2019 IEEE-APS Topical Conference on Antennas and Propagation in Wireless Communications (APWC), Granada, Spain, 9–13 September 2019; pp. 34–35.
- Brinkhoff, J.; Hornbuckle, J. Characterization of WiFi signal range for agricultural WSNs. In Proceedings of the 2017 23rd Asia-Pacific Conference on Communications: Bridging the Metropolitan and the Remote, (APCC), Perth, Australia, 11–13 December 2018; pp. 1–6.
- 74. Pan, H.; Shi, Y.; Wang, X.; Li, T. Modeling wireless sensor networks radio frequency signal loss in corn environment. *Multimed. Tools Appl.* **2017**, *76*, 19479–19490. [CrossRef]
- 75. Wu, H.; Miao, Y.; Li, F.; Zhu, L. Empirical modeling and evaluation of multi-path radio channels on wheat farmland based on communication quality. *Trans. ASABE* 2016, *59*, 759–767. [CrossRef]
- Liu, H.; Meng, Z.; Wang, M. A wireless sensor network for cropland environmental monitoring. In Proceedings of the International Conference on Networks Security, Wireless Communications and Trusted Computing (NSWCTC), Wuhan, China, 25–26 April 2009; Volume 1, pp. 65–68. [CrossRef]
- 77. Olasupo, T.O.; Alsayyari, A.; Otero, C.E.; Olasupo, K.O.; Kostanic, I. Empirical path loss models for low power wireless sensor nodes deployed on the ground in different terrains. In Proceedings of the 2017 IEEE Jordan Conference on Applied Electrical Engineering and Computing Technologies (AEECT), Aqaba, Jordan, 11–13 October 2017; pp. 1–8.
- 78. Olasupo, T.O.; Otero, C.E.; Olasupo, K.O.; Kostanic, I. Empirical path loss models for wireless sensor network deployments in short and tall natural grass environments. *IEEE Trans. Antennas Propag.* **2016**, *64*, 4012–4021. [CrossRef]
- 79. Perez, G.E.; Alsayyari, A.; Kostanic, I. Comparison of the propagation loss of a real-life wireless sensor network and its complimentary simulation model. In Proceedings of the 2015 IEEE 17th International Conference on High Performance Computing and Communications, 2015 IEEE 7th International Symposium on Cyberspace Safety and Security, and 2015 IEEE 12th International Conference on Embedded Software and Systems, New York, NY, USA, 24–26 August 2015; pp. 1832–1837.
- 80. Srisooksai, T.; Kaemarungsi, K.; Takada, J.; Saito, K. Radio propagation measurement and characterization in outdoor tall food grass agriculture field for wireless sensor network at 2.4 GHz band. *Prog. Electromagn. Res. C* 2018, *88*, 43–58. [CrossRef]
- Iswandi; Nastiti, H.T.; Praditya, I.E.; Mustika, I.W. Evaluation of XBee-Pro transmission range for Wireless Sensor Network's node under forested environments based on Received Signal Strength Indicator (*RSSI*). In Proceedings of the 2nd International Conference on Science and Technology-Computer, ICST, Yogyakarta, Indonesia, 27–28 October 2016; pp. 56–60.
- Gay-Fernández, J.A.; Cuiñas, I. Peer to peer wireless propagation measurements and path-loss modeling in vegetated environments. *IEEE Trans. Antennas Propag.* 2013, 61, 3302–3311. [CrossRef]
- Mathew, K.; Tabassum, M. Analysis of bluetooth and zigbee signal penetration and interference in foliage. In *Proceedings of the International MultiConference of Engineers and Computer Scientists (IMECS)*; Hong Kong, China, 16–18 March 2016; Volume 2, pp. 547–552.
- He, Y.; Zhang, W.; Jiang, N.; Luo, X. The research of wireless sensor network channel propagation model in the wild environment. In Proceedings of the 9th International Conference on P2P, Parallel, Grid, Cloud and Internet Computing 3PGCIC, Guangdong, China, 8–10 November 2014; pp. 227–231. [CrossRef]
- Liu, L.; Yao, Y.; Cao, Z.; Zhang, M. DeepLoRa: Learning Accurate Path Loss Model for Long Distance Links in LPWAN. In Proceedings of the IEEE INFOCOM, Virtual, 10–13 May 2021.
- Manpreet; Malhotra, J. ZigBee technology: Current status and future scope. In Proceedings of the 2015 International Conference on Computer and Computational Sciences (ICCCS), Greater Noida, India, 27–29 January 2015; pp. 163–169. [CrossRef]
- 87. Sadowski, S.; Spachos, P. Wireless technologies for smart agricultural monitoring using internet of things devices with energy harvesting capabilities. *Comput. Electron. Agric.* **2020**, *172*, 105338. [CrossRef]
- Danbatta, S.J.; Varol, A. Comparison of Zigbee, Z-Wave, Wi-Fi, and Bluetooth Wireless Technologies Used in Home Automation. In Proceedings of the 7th International Symposium on Digital Forensics and Security (ISDFS), Barcelos, Portugal, 10–12 June 2019; pp. 1–5.
- Gloria, A.; Cercas, F.; Souto, N. Comparison of communication protocols for low cost Internet of Things devices. In Proceedings of the South-East Europe Design Automation, Computer Engineering, Computer Networks and Social Media Conference, SEEDA-CECNSM, Kastoria, Greece, 23–25 September 2017.
- Peng, Y.L.; Li, P.P.; Wang, J.Z.; Hu, Y.G.; Lin, Y.F. Propagation characteristics of 2.4 GHz wireless channel at different directions and heights in tea plantation. *Appl. Mech. Mater.* 2013, 325–326, 1697–1701. [CrossRef]
- 91. Benaissa, S.; Plets, D.; Tanghe, E.; Verloock, L.; Martens, L.; Hoebeke, J.; Sonck, B.; Tuyttens, F.A.M.; Vandaele, L.; Stevens, N.; et al. Experimental characterisation of the off-body wireless channel at 2.4 GHz for dairy cows in barns and pastures. *Comput. Electron. Agric.* **2016**, *127*, 593–605. [CrossRef]
- Zhang, W.; He, Y.; Liu, F.; Miao, C.; Sun, S.; Liu, C.; Jin, J. Research on WSN channel fading model and experimental analysis in orchard environment. In *Computer and Computing Technologies in Agriculture V. CCTA 2011*; IFIP Advances in Information and Communication Technology Book Series (IFIPAICT, Volume 369); Springer: Berlin/Heidelber, Germany, 2012; pp. 326–333. [CrossRef]
- Samijayani, O.N.; Mujadin, A.; Darwis, R.; Astharini, D.; Rahmatia, S. Hybrid ZigBee and WiFi Wireless Sensor Networks for Hydroponic Monitoring. In Proceedings of the 2020 International Conference on Electrical, Communication, and Computer Engineering (ICECCE), Istanbul, Turkey, 12–13 June 2020; pp. 1–4.

- Botella-campos, M.; Parra, L.; Sendra, S.; Lloret, J. WLAN IEEE 802. 11b/g/n Coverage Study for Rural Areas. In Proceedings of the 2020 International Conference on Control, Automation and Diagnosis (ICCAD), Paris, France, 7–9 October 2020; pp. 1–6.
- Lavanya, U.; Mupparaju, S.; Patnala, P.; Anugu, P.R.; Surendran, S. Model Selection for Path Loss Prediction in Wireless Networks. In Proceedings of the 2020 International Conference on Communication and Signal Processing (ICCSP), Chennai, India, 28–30 July 2020; pp. 1490–1493.
- 96. Lora Alliance What Is LoRaWAN Specification. Available online: https://lora-alliance.org/about-lorawan/ (accessed on 1 February 2021).
- Amali, K.; Sharil, M.; Devi, J. Impact of foliage on LoRa 433MHz propagation in tropical environment Impact of Foliage on LoRa 433 MHz Propagation in Tropical Environment. In *AIP Conference Proceedings*; AIP Publishing LLC: Melville, NY, USA, 2019; Volume 20009, pp. 1–7.
- Avila-Campos, P.; Astudillo-Salinas, F.; Vazquez-Rodas, A. Evaluation of LoRaWAN Transmission Range for Wireless Sensor Networks in Riparian Forests. In Proceedings of the 22nd International ACM Conference on Modeling, Analysis and Simulation of Wireless and Mobile Systems (MSWiM), Miami, FL, USA, 25 November 2019; pp. 199–206.
- 99. Zarnescu, A.; Ungurelu, R.; Secere, M.; Varzaru, G.; Mihailescu, B. Implementing a large LoRa network for an agricultural application. In Proceedings of the 7th International Conference on Energy Efficiency and Agricultural Engineering (EE&AE), Ruse, Bulgaria, 12–14 November 2020; pp. 4–8. [CrossRef]
- Anzum, R. Factors that affect LoRa Propagation in Foliage Factors that affect LoRa Propagation Foliage Medium Factors that affect LoRa Propagation in Foliage Factors that affect LoRa Propagation Foliage Medium Factors that affect LoRa Propagation in Foliage Medium. *Procedia Comput. Sci.* 2021, 194, 149–155. [CrossRef]
- 101. Anzum, R.; Hadi Habaebi, M.; Islam, R.; Hakim, G.P.N. Modeling and Quantifying Palm Trees Foliage Loss using LoRa Radio Links for Smart Agriculture Applications. In Proceedings of the 7th International Conference on Smart Instrumentation, Measurement and Applications (ICSIMA), Bandung, Indonesia, 23–25 August 2021; pp. 105–110. [CrossRef]
- Phaiboon, S.; Phokharatkul, P. An Empirical Path Loss Model for Wireless Sensor Network Placement in Banana Plantation. In Proceedings of the Progress in Electromagnetics Research Symposium, Hangzhou, China, 21–25 November 2021; pp. 354–357.
- AlSayyari, A.; Kostanic, I.; Otero, C.E. An Empirical Path Loss Model for Wireless Sensor Network Deployment in a Dense Tree Environment. In Proceedings of the 2017 IEEE Sensors Applications Symposium (SAS), Glassboro, NJ, USA, 13–15 March 2017; pp. 1–6.
- Alsayyari, A.; Aldosary, A. Path loss results for wireless sensor network deployment in a sparse tree environment. In Proceedings of the 2019 International Symposium on Networks, Computers and Communications (ISNCC), Istanbul, Turkey, 18–20 June 2019; pp. 1–6. [CrossRef]
- 105. Xu, X.; Zhang, Z.; Xu, Y.; Yang, Z.; Chen, Y.; Liang, Z.; Zhou, J.; Zheng, J. Measurement and Analysis of Wireless propagative Model of 433MHz and 2.4GHz Frequency in Southern China Orchards. *IFAC-PapersOnLine* **2018**, *51*, 695–699. [CrossRef]
- 106. Navarro, A.; Guevara, D.; Florez, G.A. An Adjusted Propagation Model for Wireless Sensor Networks in Corn Fields. In Proceedings of the 2020 XXXIIIrd General Assembly and Scientific Symposium of the International Union of Radio Science, Rome, Italy, 29 August–5 September 2020; pp. 1–3.